Author’s response to reviews

Title: Zika virus disease, microcephaly and Guillain-Barré syndrome in Colombia: epidemiological situation during 21 months of the Zika virus outbreak, 2015-2017

Authors:

Nelson Méndez (nemendez15@gmail.com)

Misael Oviedo-Pastrana (mpastrana@outlook.com)

Salim Mattar (mattarsalim@hotmail.com)

Isaac Caicedo-Castro (isacaic@correo.unicordoba.edu.co)

German Arrieta (arrietager@yahoo.es)

Version: 3 Date: 03 Sep 2017

Author’s response to reviews:


Reviewer reports:

The two suggestions made by the reviewers were carried out, in green color the changes.

Reviewer: One abbreviation has not been defined: ZIKV

REPLY/ Introduction of the abbreviation was done, page 7 in green color

Reviewer: Please review the last sentence: “Data limitations like absence of disaggregate data by age-group and absence of disaggregate data by altitude. Such data is vital in control.” Change to
e.g. “Data essential in ZIKA control such as disaggregated data by age-group and by altitude are currently missing.”

REPLY/ We took the suggested sentence now is included. Page 24 in green color